Author's response to reviews

Title: Viral Transmission Risk Factors in an Egyptian Population with High Hepatitis C Prevalence

Authors:

Mary Kate Mohlman (m.k.mohlman@gmail.com)
Doa’a A. Saleh (doa_a_saleh@yahoo.com)
Sameera Ezzat (drsekhir@yahoo.com)
Mohamed Abdel-Hamid (vhrlathcp@yahoo.com)
Brent Korba (korbabe@georgetown.edu)
Kirti Shetty (ks359@georgetown.edu)
Sania Amr (samr@epi.umaryland.edu)
Christopher A. Loffredo (cal9@georgetown.edu)

Version: 5 Date: 24 September 2015

Author's response to reviews: see over
Response to Reviewers

Title: Viral Transmission Risk Factors in an Egyptian Population with High Hepatitis C Prevalence

Date: September 24, 2015

We thank the reviewers for their comments and suggestions. Changes in the revised manuscript are described below.

Reviewer: Julia Uhanova

Reviewer's report: The authors successfully addressed all comments raised by this reviewer. The revised manuscript is much improved and is clear and easy to follow.

Discretionary Revisions

Lines 86-87: Add comma and delete and at the end of the line 86. Then insert the phrase “risk factors for” after the comma before ... HCV transmission (line 87).

Response: Thank you for the suggestion, we have implemented this change.

Line 181: At the beginning of the sentence, substitute “In this group,” instead of “And”.

Response: Thank you for noticing this error, we have changed the sentence accordingly.

Sincerely,

Christopher Loffredo, PhD

Professor of Oncology and of Biostatistics